Author's response to reviews

Title: The Toronto Prehospital Hypertonic Resuscitation-Head Injury and Multi Organ Dysfunction Trial (TOPHR HIT) - Methods and Data Collection Tools

Authors:

Laurie J Morrison MD, MSc (morrisonl@smh.toronto.on.ca)
Sandro Rizoli PhD, MD (sandro.rizoli@sunnybrook.ca)
Brian Schwartz MD (bschwartz@socpc.ca)
Shawn Rind PhD (shawn.rhind@drdc-rrdc.goc.ca)
Merita Simitciu MD (singhs@smh.toronto.on.ca)
Tyrone Perreira MSc (perreirat@smh.toronto.on.ca)
Russell MacDonald MD (rmacdonald@ornge.ca)
Anna Trompeo MD (singhs@smh.toronto.on.ca)
Donald Stuss PhD (dstuss@rotman-baycrest.on.ca)
Sandra Black MD (sandra.black@sw.ca)
Alex Kiss PhD (alex.kiss@ices.on.ca)
Andrew Baker MD, PhD (bakera@smh.toronto.on.ca)

Version: 2 Date: 23 October 2009

Author's response to reviews: see over
October 23, 2009

Vikki Bean
Editorial Administrator
Trials
Bio Med Central Ltd.
Floor 6, 236 Gray's Inn Road
London, WC1X8HL

Dear Ms. Bean:

Re: The Toronto Prehospital Hypertonic Resuscitation – Head Injury and Multi Organ Dysfunction Trial (TOPHR HIT) – Methods and Data Collection Tools
MS: 1647819926315841.

Thank you to you and the Associate Editors for reviewing our manuscript. We agree with the comments by the Editor and would like to resubmit this manuscript as a study protocol. In addition we have included the Clinical Trials registration number for TOPHR HIT.

Thank you, once again for your comments and reviews, we look forward to hearing from you soon.

Regards,

Laurie J. Morrison
Clinician-Scientist & Director, Rescu
Robert and Dorothy Pitts Chair, Acute Care and Emergency Medicine
Professor, Division of Emergency Medicine
Department of Medicine, University of Toronto
Keenan Research Centre
Li Ka Shing Knowledge Institute
St Michael’s Hospital